Author’s response to reviews

Title: Consolidated Criteria for strengthening reporting of health research involving Indigenous Peoples: The CONSIDER statement

Authors:

tania huria (tania.huria@otago.ac.nz)
Suetonia Palmer (suetonia.palmer@otago.ac.nz)
Suzanne Pitama (suzanne.pitama@otago.ac.nz)
Lutz Beckert (Lutz.Beckert@cdhb.health.nz)
Cameron Lacey (cameron.lacey@otago.ac.nz)
Shawn Ewen (shaun.ewen@unimelb.edu.au)
Linda Tuhiwai Smith (tuhiwai@waikato.ac.nz)

Version: 2 Date: 03 Aug 2019

Author’s response to reviews:

Tania Huria

PhD candidate
Māori and Indigenous Health Institute
University of Otago Christchurch
2 Riccarton Ave
Christchurch 8140
Aotearoa New Zealand

Pentti Nieminen
BMC Medical Research Methodology

January 8, 2017

Tēnā koe

We wish to thank the editors and reviewers of BMC Medical Research Methodology for their review of "Consolidated Criteria for strengthening reporting of health research involving Indigenous Peoples: The CONSIDER statement" (BMRM-D-19-00226)
The authors of this study appreciate the time that the editors and reviewers have taken to review the manuscript and provide recommendations. We have made all of the suggested changes by the editor, these include:

1. Inserting Results heading

2. Under Funding declaration, stating if the funding body had any role in the research.

3. Clarification of Shaun Ewen's, Cameron Lacey's and Linda Tuhiwai Smiths contribution to the research under Authors Contribution.

4. Correcting the initials SGP to SCP.

We have provided a clean version of the manuscript with no track changes. We thank you again for your further consideration of the revised manuscript.

Ngā mihi

Tania Huria, on behalf of the authors